Author's response to reviews

Title: Temporal changes in haematocrit following artemisinin-based combination treatments of uncomplicated falciparum malaria in children

Authors:

Akintunde Sowunmi (akinsowunmi@hotmail.com)
Kazeem Akano (heycarnow@yahoo.com)
Adejumoke I Ayede (idayede@yahoo.com)
Godwin Ntadom (ntadomg@yahoo.com)
Bayo Fatunmbi (fatunmbib@wpro.who.int)
Temitope Aderoyeje (tope_forever@yahoo.com)
Elsie O Adewoye (elolade@yahoo.com)

Version: 6 Date: 12 August 2015

Author's response to reviews:

Department of Clinical Pharmacology
University College Hospital
Ibadan
12 August 2015

The Editor
BMC Infectious Diseases

Dear Editor,

Revised MS: 4977010871594931 - Temporal changes in haematocrit following artemisinin-based combination treatments of uncomplicated falciparum malaria in children

Thank you your comment and reminders at all times. The trials from which data reported in the current manuscript were evaluated had been registered retrospectively. Trial registration numbers of these trials had been included in the abstract and the method sections.

All changes made have been underlined.

We hope the inclusion of the trial numbers is satisfactory.

With kind regards
Akintunde Sowunmi